Author's response to reviews

Title: Homing and reparative effect of intra-articular injection of autologus mesenchymal stem cells in arthritic animal model

Authors:

Abir Mokbel (abir138@yahoo.com)  
Omar El-Tookhy (oeltookhy@gmail.com)  
Ashraf Shamaa (shamaa_61@yahoo.com)  
Laila Rashed (drlalarashed20002000@yahoo.com)  
Dina Sabry (dinnasabry69@yahoo.com)  
Abeer Mostafa (abeerelsayed9@yahoo.com)

Version: 2 Date: 20 September 2011

Author's response to reviews: see over
The title of submitted paper is:

HOMING AND REPARATIVE EFFECT OF INTRA-ARTICULAR INJECTION OF AUTOLOGOUS MESENCHYMAL STEM CELLS IN ARTHRITIC ANIMAL MODEL

I want to submit this revised corrected original research article to BMC Musculoskeletal disorders because.

The corresponding author:

Dr. Dina Sabry Abd ElFatah.

Assistant Professor of Medical Biochemistry and Molecular Biology. Faculty of Medicine. Cairo University, Egypt.

Email: dinnasabry69@yahoo.com
Second email: dinasabry69@hotmail.com
Telephone: 002-0111200200
Fax: 002-023632297